Author's response to reviews

Title: The Selective Cytotoxic Anti Cancer Properties and Proteomic Analysis of Trigonella Foenum-Graecum

Authors:

  Abdulaziz Alsemari (alsemari@kfshrc.edu.sa)
  Fahad Alkhodairy (alhkodairy@kfshrc.edu.sa)
  Ahmad Aldakan (Aldakan@kfshrc.edu.sa)
  Mai Al-Mohanna (Al-Mohanna@kfshrc.edu.sa)
  Eman Bahoush (Bahoush@kfshrc.edu.sa)
  Zakia Shinwari (Shinwari@kfshrc.edu.sa)
  Ayodele Alaiya (AAlaiya@kfshrc.edu.sa)

Version: 4 Date: 1 August 2013

Author's response to reviews: see over
2013-08-01

Tom Rowles PhD
Executive Editor
BMC Complementary and Alternative Medicine

Thank you for your recommendations. The manuscript was revised as follow:

1- The title and the abstract was changed to fit the *in vitro* work.
2- The case report was summarised and was incorporated into the background section.
3- The follow-up information regarding fenugreek administration was also included in the background section.
3- The manuscript was edited by a native English speaking colleague.

Sincerely,

The corresponding author
Abdulaziz Al-Semari, MD
Department of Neurosciences
King Faisal Specialist Hospital and Research Centre.
Riyadh 11211 PO box 3354. Saudi Arabia
alsemari@kfshrc.edu.sa